Author's response to reviews

Title: Adjudication of Etiology of Acute Kidney Injury: Experience from the TRIBE-AKI multi-center study

Authors:

Jay L. Koyner (jkoyner@uchicago.edu)
Amit X Garg (Amit.Garg@lhsc.on.ca)
Heather Thiessen-Philbrook (Heather.ThiessenPhilbrook@lhsc.on.ca)
Steven G Coca (steven.coca@yale.edu)
Lloyd G Cantley (Lloyd.cantley@yale.edu)
Aldo Peixoto (Aldo.peixoto@yale.edu)
Cary S Passik (Cary.Passik@danhosp.org)
Kwangik Hong (kwangik.hong@yale.edu)
Chirag R Parikh (chirag.parikh@yale.edu)

Version: 4
Date: 28 May 2014

Author's response to reviews: see over
Dear Dr. Tolwani,

Thank you very much for the opportunity to submit our revised manuscript entitled “Adjudication of Etiology of Acute Kidney Injury: Experience from the TRIBE-AKI multi-center study” to BMC Nephrology. Our re-submission includes all of the changes requested by the editor including naming all of the individual review ethical committees, including the STROBE guidelines (now attached), and the changes to the Acknowledgements and Author's contribution sections. We feel the re-submission remains of interested to the readers of BMC Nephrology and thank you for the opportunity to publish our work in your journal.

Sincerely,

Jay Koyner (on behalf of all TRIBE-AKI investigators)